Author’s response to reviews

Title: Urgent referral for suspected cancer as a means of reducing delay in the diagnosis. A population-based observational study in Denmark

Authors:

Mette B Larsen (mette.bach.larsen@alm.au.dk)
Rikke P Hansen (rph@alm.au.dk)
Dorte G Hansen (dgilsaa@health.sdu.dk)
Frede Olesen (fo@alm.au.dk)
Peter Vedsted (p.vedsted@alm.au.dk)

Version: 4 Date: 9 January 2013

Author’s response to reviews:

I have added the abstract to the manuscript.